\begin{table}%t1
\par
\caption {J2000 coordinates and cross-references for candidate stars.
\label{ids}}
%\centerline
 {\begin{tabular}{rllrrrrrrl}
\hline\hline%\noalign{\smallskip}
 No. &  \multicolumn{1}{c}{RA} & \multicolumn{1}{c}{Dec} & Malm  & Arty & Upgren & Argue & Abad & \multicolumn{1}{c}{BD} & Other \\
\hline%\noalign{\smallskip}
203 & 12 19 16.26 & +25 26 10.5 & 25.087 &   44 & 26.082 &       & 1038 &                 & Tr A3 \\
213 & 12 24 05.72 & +26 07 43.1 & 26.170 &      &        &       & 1431 & +26$\degr$ 2342 & Tr A13 \\
214 & 12 24 17.15 & +24 19 28.3 & 24.148 &  349 &        &       & 1444 & +25$\degr$ 2502 & Tr A14, HIP 60~511, J122417.15+241928.4 \\
219 & 12 27 52.66 & +25 05 44.7 & 25.198 &  603 & 25.073 &       & 1748 &                 & Tr A19 \\
220 & 12 28 56.43 & +26 32 57.2 & 27.218 &  648 & 27.108 &       & 1826 & +27$\degr$ 2139 & Tr A20, J122856.43+263257.4 \\  
221 & 12 30 14.10 & +25 01 42.3 & 25.230 &      & 25.078 &       & 1924 &                 & Tr A21, HIP 61~012 \\
222 & 12 20 05.71 & +25 54 36.8 & 26.101 &   72 & 26.087 & 11--14 & 1097 &                 & Tr B1  \\
275 & 12 18 51.03 & +26 19 24.4 & 26.075 &   20 & 27.084 &  8--21 &  995 & +27$\degr$ 2113 & \\
280 & 12 19 05.18 & +26 11 04.4 & 26.084 &      & 26.080 &  9--10 & 1019 &                 & \\  
282 & 12 21 15.61 & +26 09 14.0 & 26.121 &  132 & 26.092 &  9--16 & 1183 &                 & J122115.63+260914.1 \\
293 & 12 18 58.70 & +26 03 30.9 & 26.079 &      & 26.077 & 10--21 & 1009 &                 & \\  
299 & 12 24 22.95 & +25 54 56.2 & 26.175 &      &        & 11--24 & 1453 &                 & \\  
311 & 12 22 40.65 & +25 40 11.9 & 26.141 &      &        & 13--18 & 1297 &                 & \\  
334 & 12 22 25.82 & +25 19 03.3 & 25.129 &      & 26.093 & 15--16 & 1278 &                 & \\  
351 & 12 19 08.52 & +25 03 11.6 & 25.084 &      & 25.053 & 17--14 & 1023 &                 & \\  
379 & 12 27 26.28 & +24 46 57.0 & 25.194 &  577 & 25.071 & 19--28 &      &                 & \\
386 & 12 13 43.90 & +22 53 16.7 & 23.039 &      &        &       &  589 & +23$\degr$ 2433 & HD 106~293, Bou 38 \\
390 & 12 52 11.61 & +25 22 24.6 & 25.449 &      &        &       &      & +26$\degr$ 2402 & HD 111~878, HIP 62~805 \\
395 & 12 32 31.07 & +35 19 52.3 & 35.177 &      &        &       &      & +36$\degr$ 2278 & 
HIP 61~205, Bou 50 \\
399 & 12 51 41.92 & +27 32 26.5 & 28.448 &      &        &       &      & +28$\degr$ 2156 & HD 111~812, HIP 62~763, LS Com \\
400 & 13 00 35.18 & +23 39 06.3 &        &      &        &       &      & +24$\degr$ 2522 & HD 113~037, HIP 63~493 \\
414 & 12 34 46.91 & +24 09 37.9 & 24.292 &  993 &        &       & 2248 &                 & J123446.93+240937.7 \\ 
415 & 12 36 52.49 & +24 01 27.1 & 24.314 & 1053 &        &       &      &                 & \\ 
416 & 12 16 49.44 & +24 51 21.1 & 25.055 &      & 25.044 &       &  833 & +25$\degr$ 2479 & \\ 
417 & 12 22 59.42 & +24 58 58.4 & 25.141 &  245 & 25.058 &       & 1332 &                 & \\ 
418 & 12 18 57.26 & +25 53 10.9 & 26.078 &   23 & 26.076 &       & 1003 &                 & \\ 
420 & 12 23 28.17 & +25 53 38.8 & 26.155 &  278 &        &       &      &                 & J122328.21+255339.9 \\      
421 & 12 26 51.04 & +26 16 02.5 & 26.215 &  537 & 27.101 &  8--06 & 1660 &                 & J122651.03+261601.9 \\ 
422 & 12 29 44.66 & +25 32 35.5 & 26.250 &  686 & 26.107 &       & 1895 &                 & \\
423 & 12 33 30.19 & +26 10 00.2 & 26.286 &  908 & 26.115 &       & 2166 &                 & J123330.19+261000.1 \\
424 & 12 38 55.36 & +25 53 38.1 & 26.364 &      & 26.126 &       &      &                 & \\ 
425 & 12 41 04.97 & +25 42 17.9 & 26.392 &      & 26.130 &       & 2406 & +26$\degr$ 2379 & \\ 
427 & 12 33 54.22 & +27 08 04.7 & 27.297 &  938 & 27.121 &       & 2194 &                 & \\
428 & 12 51 25.12 & +26 47 08.7 & 27.506 &      &        &       &      &                 & \\       
430 & 12 18 39.65 & +27 45 58.8 & 28.108 &  101 & 28.068 &       &      &                 & \\ 
431 & 12 33 00.68 & +27 42 45.1 & 28.277 &  877 & 28.096 &       & 2129 &                 & J123300.62+274244.8 \\ 
432 & 12 39 45.90 & +28 17 56.7 & 28.345 &      & 29.129 &       &      &                 & \\
433 & 12 40 13.00 & +28 13 21.0 & 28.351 &      &        &       & 2395 & +29$\degr$ 2313 & HD 110~195 \\ 
434 & 12 28 25.56 & +29 53 54.4 & 30.178 &      & 30.077 &       &      & +30$\degr$ 2276 & \\ 
435 & 12 28 27.51 & +29 52 42.3 & 30.179 &      & 30.078 &       &      & +30$\degr$ 2277 & \\ 
436 & 12 35 00.33 & +30 11 33.6 & 30.252 &      &        &       &      & +30$\degr$ 2296 & HIP 61~409 \\
437 & 12 21 20.91 & +32 10 35.2 & 32.078 &      &        &       &      & +32$\degr$ 2236 & HD 107~512, HIP 60~256, Bou 49 \\ 
%\noalign{\smallskip}
\hline
\end{tabular}}
\par
%\smallskip
Note: this table displays the star numbers according to the numbering system used in WEBDA (\url{http://www.univie.ac.at/webda/})
or extends it (No~$>$~413), J2000 coordinates from SIMBAD or 2MASS, cross-references with the lists of 
Malmquist (\cite{malm1},~\cite{malm2}), Artjukhina (\cite{art55}), Upgren (\cite{up62}), Argue (\cite{arg63}) and 
Abad \& Vicente (\cite{av99}). BD refers, as usual, to the Bonner Durchmusterung. 
Tr~A3, A13, A14, A19, A20, A21 and Tr~B1 refer to Trumpler's (\cite{tr38}) appendices A and B respectively. J designation are from Casewell et~al. (\cite{cjd06}). ``Bou'' refers to Bounatiro (\cite{lb93}). 
Miscellanous identifications are: HD for the Henry Draper catalogue and HIP for the HIPPARCOS catalogue.
\end{table}